Physics 606 Techniques of Computational Physics

and other background materials will be provided as the course developes.

Course Instructor:

Comer Duncan Telephone: 372 8108; you may also visit my web page.

Course Information Course Materials:

Hand-outs on various computational physics methods will be provided. the term proceeds.

A bibliography of some key references will be provided.

Course Objectives:

Unix Some Numerical Methods for Ordinary differential equations Hyperbolic Partial Differential Equations Introdution to Computational Fluid Dynamics Introduction to Parallel Computing

computing applications Introduction to Cluster Computing

good treatment of which you should consult when needed.

the students and instructor.

Tentative Course Syllabus for Fall Semester 2002

Topic Weeks Notes

Introduction to Unix 1 week

1 week

Introduction to Fortran 90 2 weeks 2 weeks

1 week

Conditions in f90

1 week

worked applications) 2 weeks

equation in f90 2 weeks 2 weeks dynamical systems 2 weeks

Requirements for the course:

Examinations:

exercises] as 50% and the term project as %50.